

# **Geophysical Research Letters**

# **RESEARCH LETTER**

10.1029/2019GL081967

#### **Key Points:**

- Statistical models to describe extreme river discharge can be unreliable when multiple processes lead to extreme events
- Large ensemble model experiments (many simulation years) are suitable for analysis of extreme events and do not rely on statistical models
- Hydrological large ensembles provide better estimates of changes in extreme floods and droughts and their characteristics

#### **Supporting Information:**

Supporting Information S1

#### Correspondence to:

K. van der Wiel, wiel@knmi.nl

#### **Citation**:

Van der Wiel, K., Wanders, N., Selten, F. M., & Bierkens, M. F. P. (2019). Added value of large ensemble simulations for assessing extreme river discharge in a 2 °C warmer world. *Geophysical Research Letters*, 46, 2093–2102. https://doi.org/10.1029/2019GL081967

Received 10 JAN 2019 Accepted 3 FEB 2019 Accepted article online 7 FEB 2019 Published online 18 FEB 2019

#### ©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Added Value of Large Ensemble Simulations for Assessing Extreme River Discharge in a 2 °C Warmer World

#### K. van der Wiel<sup>1</sup>, N. Wanders<sup>2,3</sup>, F. M. Selten<sup>1</sup>, and M. F. P. Bierkens<sup>2,4</sup>

<sup>1</sup>Royal Netherlands Meteorological Institute, De Bilt, Netherlands, <sup>2</sup>Geosciences, Physical Geography, Utrecht University, Utrecht, Netherlands, <sup>3</sup>Department of Civil and Environmental Engineering, Princeton University, Princeton, NJ, USA, <sup>4</sup>Deltares, Utrecht, Netherlands

**Abstract** The assessment of return periods of extreme hydrological events often relies on statistical analysis using generalized extreme value (GEV) distributions. Here we compare the traditional GEV approach with a novel large ensemble approach to determine the added value of a direct, empirical distribution-based estimate of extreme hydrological events. Using the global climate and hydrological models EC-Earth and PCR-GLOBWB, we simulate 2,000 years of global hydrology for a present-day and 2 °C warmer climate. We show that the GEV method has inherent limitations in estimating changes in hydrological extremes, especially for compound hydrological events. The large ensemble method does not suffer from these limitations and quantifies the impacts of climate change with greater precision. The explicit simulation of extreme events enables better hydrological process understanding. We conclude that future studies focusing on the impact of climatic changes on hydrological extremes should use large ensemble techniques to properly account for these rare hydrological events.

**Plain Language Summary** Extreme hydrological events such as droughts and floods can cause severe harm to people and nature. It is therefore important to understand why and how often they occur now and in the future. We compare two methods of studying these extreme events: a frequently used statistical method and a new direct simulation method (called "large ensemble simulations"). We show that this new method better represents the extreme events, that it reduces the uncertainties of the expected effects of climate changes on extreme events, and that it allows us to study why extreme events occur. We therefore are better capable to quantify the impact of climate change on hydrological extremes, and we recommend the large ensemble method for future studies on extreme events.

# 1. Introduction

Global climate change affects the hydrological cycle around the world. Projected changes include changes in climatological precipitation patterns, precipitation types, evaporation amounts, soil moisture availability, and discharge levels (e.g., Berghuijs et al., 2014; Held & Soden, 2006; Intergovernmental Panel on Climate Change, 2013; Milly et al., 2005; Wanders & Wada, 2015a). Besides changes in mean conditions, also extreme hydrological events are projected to change, including more frequent extreme precipitation events, more severe droughts, and changing flood occurrence (e.g., Min et al., 2011; Prudhomme et al., 2014; Van der Wiel et al., 2017; Wanders & Van Lanen, 2015; Wanders & Wada, 2015b; Winsemius et al., 2015). These extreme events can have severe negative impacts on societies and ecosystems. For instance, the recent drought in South Africa (2015 to present, Baudoin et al., 2017) and the 2017 floods in Bangladesh (Philip et al., 2018) had severe negative consequences on people's livelihoods, agricultural production, and the national economy. Scientific understanding of the probability of occurrence, severity, and characteristics of extreme hydrological events is therefore of societal importance.

Extreme events can be studied in two ways: (i) based on long time series and sampling or (ii) by means of a statistical model of the tail of the distribution. The limited length of observational hydrological records means we frequently rely on the second approach using a statistical model for statements regarding extreme hydrological events and also for simulation studies such an approach is frequently taken (e.g., Abaurrea & Cebrián, 2002; Smith et al., 2014; Sousa et al., 2011). These statistical models are used to extrapolate sample statistics about the extreme values of limited-length observational records, to provide estimates of the



probabilities of unobserved extreme values. It is assumed that extreme values follow a single given probability distribution. This is common practice when estimating defense levels against floods and designing drought protection infrastructure (e.g., dams and reservoirs).

Recent advancements of computing power now allow for sufficiently long model simulations or large ensembles to be created to study extreme events by means of sampling, therewith removing the need to rely on statistical models, assumptions on distributions, and extrapolation of data. This is especially valuable when looking at hydrological extremes, which are sensitive to minor changes in the parameters of statistical models and their extrapolations (Engeland et al., 2004), often leading to great uncertainties about extreme event estimates. Of course, a modeling approach depends on the quality of modeled data. Model validation with observations remains important.

Here we aim to contribute to this problem by showing that a large ensemble modeling approach can improve estimates of the risk of extreme events in the present climate and reduce uncertainties in the future projections of the occurrence of these extreme events. We use output from a global climate model (GCM) to force a global hydrological model (GHM) and create two large ensembles of 2,000 years each. Taking the perfect model approach, we show the advantages of the large ensemble approach and the explicit simulation of extreme events over the common statistical approach. We provide global estimates of changes in extreme river discharge that may occur in a world with 2 °C global mean warming, both for extreme low discharge (droughts) and extreme high discharge (floods). We mostly focus on extreme events with an average return period of 100 years, which is representative of flood protection levels around the world (Ward et al., 2017) and sampled relatively well in these simulations.

### 2. Models and Methods

#### 2.1. Global Climate Model: EC-Earth v2.3

The EC-Earth GCM (Hazeleger et al., 2012) was used to create two large ensembles of meteorological data. We used the model in the same configuration (v2.3, 1.1° resolution) as in the Coupled Model Intercomparison Project phase 5 (CMIP5, Taylor et al., 2012). EC-Earth combines an atmospheric model, an ocean model, a land surface model, and a sea ice model. Full details on EC-Earth, its configuration, parametrizations, and individual components are provided in Hazeleger et al. (2012).

The two large ensembles represent two periods which differ in global mean surface temperature (GMST, supporting information Figure S1; James et al., 2017). The first "present-day" large ensemble was created to have a modeled absolute GMST equal to observed GMST in the years 2011–2015 (based on HadCRUT4 data; Morice et al., 2012). The second "2 °C warming" large ensemble was created to have a modeled absolute GMST equal to observed preindustrial temperature +2 °C warming, motivated by the Paris climate agreements (United Nations Framework Convention on Climate Change, 2015). Each large ensemble consists of 2,000 years of global, daily meteorological data. More details on the design of the large ensembles are provided in the supporting information (Buizza et al., 1999).

#### 2.2. Global Hydrological Model : PCR-GLOBWB 2

Terrestrial hydrology was simulated using the GHM PCR-GLOBWB 2 (Sutanudjaja et al., 2018). PCR-GLOBWB computes a water balance at a 0.5° grid scale, forced by daily precipitation, 2-m air temperature, and potential evapotranspiration from EC-Earth. Potential evapotranspiration was calculated from EC-Earth variables following the Penman-Monteith procedure (Zotarelli et al., 2010). PCR-GLOBWB includes modules to simulate land surface processes, groundwater, irrigation, domestic and industrial water use, and river routing including lakes and man-made reservoirs. Full details are provided in Sutanudjaja et al. (2018). We compare simulated extreme discharge events to those in the Global Runoff Data Centre data set.

A bias correction was applied to the EC-Earth variables before they were used as meteorological drivers for PCR-GLOBWB. Regridding from the coarser EC-Earth grid to the PCR-GLOBWB grid was done by means of a bilinear interpolation. Near-surface temperatures and potential evapotranspiration were corrected to have a seasonal cycle and interannual variability as observed by a shift of the mean and a multiplicative correction. Precipitation was corrected to have the same number of monthly wet days and monthly totals as observed. This is essential because excessive drizzle in EC-Earth (a common problem in GCMs; Dai, 2006) otherwise prevents realistic droughts developing in the simulations. Correction parameters were derived by



**Figure 1.** Changes in extreme floods in different rivers, assessed using three methods. (left column) GEV fit to 100 years of data, (middle column) GEV fit to 2,000 years of data, and (right column) empirical distribution estimate based on 2,000 years of data. (a-c) Lena, (d-f) Mekong, (g-i) Columbia, and (j-l) Amazon. Black lines show the present-day climate, red lines show the 2 °C warming projection, shading shows a 95% confidence interval, and green crosses show observed data. GEV = generalized extreme value.





**Figure 2.** Maps for four (a) 20-year and (b) 500-year flood events in the Amazon River. Shown is normalized river discharge (%), normalization based on the discharge level of the typical 20/500-year event at the river mouth.

comparison of the simulated present-day climate and an observational-based baseline (ERA-Interim reanalysis product, 1979–2016; Dee et al., 2011). The obtained correction factors are applied to both the present-day and 2 °C warming ensembles assuming that biases in EC-Earth are equal for the two climate states.

#### 2.3. Analysis of Extreme Events

For each land point with an annual mean discharge exceeding  $10 \text{ m}^3$ /s in the present-day climate, we computed the annual minimum and maximum discharge value. This data set was then used to analyze return levels of extreme events using three methods:

- 1. Generalized extreme value (GEV) fit to 100 years of data
- 2. GEV fit to 2,000 years of data
- 3. Empirical distribution estimate based on 2,000 years of data

The first two methods follow a commonly used statistical approach: GEV estimation. We fit GEVs and estimate confidence intervals to a randomly selected subsample of 100 years and to the full ensemble of 2,000 years (Gilleland & Katz, 2016). The first approach somewhat mimics the reality in operational water management with limited observational records or short model experiments. Comparing the two GEV methods shows the benefits of having long modeled time series.

The third method does not rely on an a priori assumption of the statistical distribution of the data. GEV fits are made under the assumption that the full distribution of extreme events can be described by a GEV function. This assumption is not always valid, especially for rivers where different runoff generating processes can cause extreme events. In that case, extreme events may not be part of a single statistical distribution, and a GEV analysis will be unreliable. We used the full empirical distribution of events as simulated in the large ensembles and used it to directly determine discharge levels for different return periods. Confidence intervals are estimated by means of bootstrap resampling (N = 10,000). Comparing the GEV fit based on 2,000 years of data to this empirical distribution approach shows the benefits of direct sampling over a statistical approach. A comparison of the empirical distribution approach to fits based on a log-normal or Gumbel distributions is included in supporting information Figures S2 and S3). The GEV-based fits provide better statistical descriptions of the modeled data than the fits based on the other distributions, we therefore only show the GEV-based analysis here.

#### 3. Results

#### 3.1. Changes in Extreme Floods

We compare the three methods of analyzing extreme flood events for some of the major rivers around the world (Figure 1). These rivers were selected to be highlighted because they display different behavior in response to global climate change and for the different methods. The model reproduces floods in these rivers quite well (Figures 1c, 1f, 1i, and 1l). However, the limited length of observational records prevents the verification of the most extreme simulated floods.



# **Geophysical Research Letters**



**Figure 3.** Maps showing statistically significant discharge changes in 100-year flood events (%). (a) Empirical distribution estimate based on 2,000 years of data, (b) GEV fit to 100 years of data, and (c) GEV fit to 2,000 years of data. In the boxes the total land fraction with significant/nonsignificant negative/positive change are noted. GEV = generalized extreme value.

Uncertainties in the GEV fit to 100 years of data are so large that we find no significant change in the flood occurrence for the Lena River (Figure 1a). The GEV fit to 2,000 years shows a statistically significant increase in flood magnitudes (Figure 1b). Based on the empirical distribution approach, we confirm this increase in flood magnitude for all return periods (Figure 1c). The GEV-fitted estimates of the 100-year flood magnitude are 4% and 1% lower for the GEV fits based on 100 and 2,000 years, respectively, than those based on the empirical distribution (supporting information Figure S4).

For the Mekong River, the GEV fit to 100 years of data again does not show significant change (Figure 1d), when using 2,000 years of data to fit the GEV (Figure 1e) a statistically significant increase of floods of all return periods is found. The GEV estimates of the 100-year flood are 8% and 1% lower than the empirical estimates (supporting information Figure S4). The empirical distribution approach here indicates that only floods with return periods of up to 200 years are projected to increase; changes of more extreme floods are uncertain (Figure 1f).

In contrast to the Lena and Mekong Rivers, no changes in flood occurrence are projected in the Columbia River based on all three methods (Figures 1g–1i). The Columbia River shows a shift in the distribution for the most extreme floods, caused by cooccurrence of severe precipitation, high snowmelt volumes, and high groundwater levels. This behavior is not observed in the 2 °C warming climate, likely due to a reduction of the contribution of snowmelt (Vano et al., 2015) or due to the fact that a temporal shift of the snowmelt season reduces cooccurrence of these processes.

The GEV fit to 2,000 years of data fails to capture the most extreme events in the Amazon River (Figure 1k). The GEV-based estimates of the 100-year flood are 8% lower than the empirical distribution for both GEV fits (supporting information Figure S4), indicating that the use of longer time series in the GEV approach does not help here. Larger differences are found for greater return periods. In this region we see that the







cooccurrence of severe precipitation in different contributing regions of the Amazon causes the most severe floods in the downstream regions of the river (as in 2014; Espinoza et al., 2014). These very extreme floods are part of a different distribution than the distribution of less extreme floods. Any statements on discharge levels of extreme floods or projected changes based on these GEV fits are unreliable. The empirical distribution approach, which is capable of capturing the double distribution in extreme floods and indeed shows multiple branches in the southeast catchment with very high discharge for very extreme floods (Figure 2), shows no projected change in extreme floods in the Amazon River (Figure 11).

We apply the same three methods to investigate changes in the value of the 1-in-100-year flood globally (Figure 3). In these maps, we only show statistically significant changes to highlight differences in statistical significance between methods. Based on the empirical distribution method, we find that 25% of land



# **Geophysical Research Letters**





shows a significant increase of 1-in-100-year floods (Figure 3a). Just 5% of global land shows a significant decrease in 1-in-100-year floods. The remaining land surface area mostly shows nonsignificant increases in 1-in-100-year floods. This global shift toward more extreme flooding conditions may be related to projected increases in extreme precipitation (e.g., Min et al., 2011; Van der Wiel et al., 2017), while at high latitudes changing precipitation types may contribute (Bintanja & Andry, 2017).

The 100-year-based GEV fit does not provide any statistically significant estimates of changes in extreme floods (Figure 3b). The large-scale pattern of changes estimated by means of the GEV fit to 2,000 years is comparable to that based on the empirical distribution approach (23% of land with significant worsening floods). Regionally differences exist; for example, in West Africa, India, and along the Nile the empirical distribution method provides more statistical significance.

#### 3.2. Changes in Extreme Droughts

We repeat the above analysis for extreme drought events. The GEV fits for extreme droughts have smaller confidence intervals (Figure 4) than those for extreme floods (Figure 1). Based on the comparison of the GEV fits to 2,000 years and the large ensemble bootstrap resampling method based on 2,000 years, we suspect the GEV fits are overconfident. Droughts in the Lena and Columbia Rivers are simulated quite well, though again very extreme droughts can not be constrained based on observational data (Figures 4c and 4i). For the Amazon River, the observed record suggests drought events are best described by a double GEV distribution, which is, however, not found for the simulated values (Figure 4l). The observational record for the Mekong River is relatively short (34 years), with one event well outside the simulated range. This one event can either be a very extreme event, more extreme than 34 years, or it is the first event of a second distribution. The limited length of the record does not allow such conclusions, nor can we say much about the quality of the model chain for very extreme events here.

The Lena River is the only river shown here, which shows a decreasing trend in extreme drought magnitude (Figures 4a–4c). The GEV fits (Figures 4a and 4b) have difficulties capturing the higher return levels,

which are caused by prolonged periods of below 0  $^{\circ}$ C temperature, below normal precipitation, and low groundwater levels. These are typically multiyear droughts, where two consecutive drought events cause a strong reduction in river discharge. These multiyear events violate the GEV assumption that events are independent, which could be the reason for the failing fits here.

The benefits of having 2,000 years of data over 100 years of data are most obvious for the Amazon River, which shows a significant increase of droughts only when the full large ensemble is taken into account (Figures 4j and 4k), and for the Mekong River, where the GEV fit to 100 years of data leads to a different conclusion than the GEV fit to 2,000 years of data (Figures 4d and 4e). In general, the difference between the GEV fits and the empirical distribution is smaller for the 2,000-year-based fits (supporting information Figure S5).

The benefits of the empirical distribution method over the GEV method are largest for the Mekong River. The GEV fit to 2,000 years does not provide a good description of the most extreme drought events (Figure 4e and supporting information Figure S5). As a result, the extreme discharge droughts are underestimated in the GEV fit compared to the values from the empirical distribution approach (Figure 4f).

Figure 5a shows the projected change for 100-year droughts following the large ensemble approach. The large-scale pattern reflects changes in large-scale precipitation patterns. For example, in Europe, a northward shift of summer precipitation (Intergovernmental Panel on Climate Change, 2013) leads to more severe 100-year droughts in the Mediterranean region and central Europe (Rhône, -20%, Danube -12%, and Rhine -14%), though less severe droughts in Scandinavia. Rivers located in the Boreal North show a strong reduction in drought severity, here global warming causes a shortening of the low flow season, reducing drought magnitudes. In equatorial Africa the widening of the Hadley circulation (Hu & Fu, 2007; Lu et al., 2007) results in worsening of droughts magnitude in the subtropical dry zone (e.g., Nile, -8%).

The 2,000-year GEV-based estimates of 100-year drought events (Figure 5c) shows comparable large-scale patterns though with less statistical significance. At smaller scales there are differences, for example, in New Guinea island where the large ensemble method indicates more severe droughts and the GEV methods show no significant change. Comparing the two 2,000-year-based methods (Figures 5a and 5c), we find that more rivers show statistically significant changes for the empirical distribution method (41% versus 28% of land surface, respectively), despite the suspected overconfidence of GEV drought estimates (Figure 4).

# 4. Discussion

The large ensembles as simulated here represent a chosen climate state. This is fundamentally different from observational records (which are subject to changing management conditions and transient climate) and transient model simulations. Analysis of transient time series requires assumptions to be made on driving covariates, for example, GMST. The design of the large ensembles used here, in combination with the empirical distribution approach, limits the need for (statistical) assumptions, descriptions, and corrections. However, for analysis of events of very high return periods sampling uncertainties remain large.

Although the hydrological model has been extensively validated and shows good agreement with observational records (Sutanudjaja et al., 2018), there is insufficient observational data to evaluate the performance of PCR-GLOBWB for extreme floods and droughts. Repeating the analysis with different GCMs and different GHMs will provide insight into the model sensitivity of the results. This is important as both EC-Earth and PCR-GLOBWB are imperfect models: coarse model resolution, model parameterizations, and missing physical processes give rise to biases in modeled fields. The present analysis was done under the perfect model assumption. This assumption can be relaxed in a multimodel framework, because the results would no longer depend on a single imperfect model formulation and biases tend to reduce (Tebaldi & Knutti, 2007). When looking at relative changes, as done here, model biases also partly cancel. Finally, by combining multiple GCMs with multiple GHMs, the dominant sources of uncertainty in projections (meteorological variability, hydrological response, or model formulation) can be isolated (e.g., Marx et al., 2018; Thober et al., 2017).

Though having an off-line GHM provides us with the opportunity to apply a bias correction to the meteorological forcing variables, there is no closure of the water budget in the complete land-atmosphere system (water budget closure is ensured in the GCM and GHM separately). The lack of full water system closure may introduce biases in future projections of drought, most notably due to the off-line computation of potential

19448007, 2019, 4, Downloaded from https://agupubs

onlinelibrary. wiley.com/doi/10.1029/2019GL081967 by Utrecht University Library, Wiley Online Library on [18/10/2023]. See the Terms

evapotranspiration (Kay et al., 2018; Milly & Dunne, 2017). Hydrological variables from GCMs do not suffer from this bias. Direct analysis of GCM hydrology is therefore a possible solution (e.g., Van der Wiel et al., 2018). However, representation of hydrological processes in GCMs is not as sophisticated as in GHMs, often neglecting the groundwater system, human-water interaction, and river routing (Bierkens, 2015). This creates a need for the GCM-GHM modeling sequence. Here we have assumed that the uncertainties introduced by the coupling are compensated by a more realistic hydrologic representation of the system.

# 5. Conclusions

The aim of this analysis was to show the added value of large ensemble GCM-GHM simulations for the study of extreme hydrological events. This novel method complements existing methods based on observational data, statistical extrapolation, and short (multi)model simulations. By means of a technical comparison we have identified limitations of GEV approximations and shown that these are resolved when extreme events are sampled directly from the empirical distribution of river discharge. The three main advantages are (i) improved estimates of discharge levels of extreme events, (ii) better constraints on projections of changes in extreme events due to global climate change, and (iii) information on the physics of extreme events.

Increasing computing power now allows us to simulate sufficiently long time series or large ensembles to sample extreme events directly from their empirical distribution. We have shown that the described limitations of GEV estimates do not apply to the large ensemble approach, assuming the model simulates extreme events and change therein with some accuracy. We therefore strongly recommend the use of large ensemble techniques for future studies of extreme hydrological events and note that similar advances can be made in studies of climate-induced extreme events in other fields.

# References

Abaurrea, J., & Cebrián, A. C. (2002). Drought analysis based on a cluster Poisson model: Distribution of the most severe drought. Climate Research, 22(3), 227-235. https://doi.org/10.3354/cr022227

- Baudoin, M.-A., Vogel, C., Nortje, K., & Naik, M. (2017). Living with drought in South Africa: Lessons learnt from the recent El Niño drought period. International Journal of Disaster Risk Reduction, 23, 128-137. https://doi.org/10.1016/j.ijdrr.2017.05.005
- Berghuijs, W., Woods, R., & Hrachowitz, M. (2014). A precipitation shift from snow towards rain leads to a decrease in streamflow. Nature Climate Change, 4(7), 583-586, https://doi.org/10.1038/nclimate2246
- Bierkens, M. F. P. (2015). Global hydrology 2015: State, trends, and directions. Water Resources Research, 51, 4923–4947. https://doi.org/ 10.1002/2015WR017173
- Bintanja, R., & Andry, O. (2017). Towards a rain-dominated arctic. Nature Climate Change, 7(4), 263-267. https://doi.org/10.1038/nclimate3240
- Buizza, R., Milleer, M., & Palmer, T. (1999). Stochastic representation of model uncertainties in the ECMWF ensemble prediction system. Quarterly Journal of the Royal Meteorological Society, 125(560), 2887-2908. https://doi.org/10.1002/qj.49712556006

Dai, A. (2006). Precipitation characteristics in eighteen coupled climate models. Journal of Climate, 19(18), 4605-4630. https://doi.org/10.1175/JCLI3884.1

Dee, D. P., Uppala, S., Simmons, A., Berrisford, P., Poli, P., Kobayashi, S., et al. (2011). The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. Quarterly Journal of the royal meteorological society, 137(656), 553-597. https://doi.org/10.1002/qj.828

Engeland, K., Hisdal, H., & Frigessi, A. (2004). Practical extreme value modelling of hydrological floods and droughts: A case study. Extremes, 7(1), 5-30. https://doi.org/10.1007/s10687-004-4727-5

Espinoza, J. C., Marengo, J. A., Ronchail, J., Carpio, J. M., Flores, L. N., & Guyot, J. L. (2014). The extreme 2014 flood in south-western Amazon basin: the role of tropical-subtropical South Atlantic SST gradient. Environmental Research Letters, 9(12), 124007. https://doi.org/10.1088/1748-9326/9/12/124007

Gilleland, E., & Katz, R. W. (2016). extRemes 2.0: An extreme value analysis package in R. Journal of Statistical Software, 72(8), 1-39.

Hazeleger, W., Wang, X., Severijns, C., Ștefănescu, S., Bintanja, R., Sterl, A., et al. (2012). EC-Earth V2.2: Description and validation of a

new seamless Earth system prediction model. Climate Dynamics, 39(11), 2611-2629. https://doi.org/10.1007/s10687-004-4727-5 Held, I. M., & Soden, B. J. (2006). Robust responses of the hydrological cycle to global warming. Journal of Climate, 19(21), 5686-5699. https://doi.org/10.1175/JCLI3990.1

Hu, Y., & Fu, Q. (2007). Observed poleward expansion of the Hadley circulation since 1979. Atmospheric Chemistry and Physics, 7(19), 5229-5236. https://doi.org/10.5194/acp-7-5229-2007

- Intergovernmental Panel on Climate Change (2013). Climate change 2013: The physical science basis. In T.F. Stocker, et al. (Eds.), Contribution of working group I to the Fifth Assessment Report of the Intergovernmental Panel on Climate Change (pp. 1535). Cambridge, UK and New York, NY: Cambridge University Press. https://doi.org/10.1017/CBO9781107415324
- James, R., Washington, R., Schleussner, C.-F., Rogelj, J., & Conway, D. (2017). Characterizing half-a-degree difference: A review of methods for identifying regional climate responses to global warming targets. Wiley Interdisciplinary Reviews: Climate Change, 8(2), e457. https://doi.org/10.1002/wcc.457
- Kay, A. L., Bell, V., Guillod, B. P., Jones, R. G., & Rudd, A. C. (2018). National-scale analysis of low flow frequency: Historical trends and potential future changes. Climatic Change, 147(3-4), 585-599. https://doi.org/10.1007/s10584-018-2145-y
- Lu, J., Vecchi, G. A., & Reichler, T. (2007). Expansion of the Hadley cell under global warming. Geophysical Research Letters, 34, L06805. https://doi.org/10.1029/2006GL028443

This work is part of the HiWAVES3 project (NWO ALWCL.2016.2). N. W. acknowledges funding by NWO 016.Veni.181.049. The authors would like to thank two anonymous reviewers and the Editor, Valeriy Ivanov, for their comments, which helped to improve the manuscript. The data set of simulated annual minimum and maximum discharge values used in this study is available from the website http://doi.org/10.5281/zenodo.2536396.

and-conditions) on Wiley Online Library for rules of use; OA articles

and Conditions (https://onlinelibrary.wiley

- Marx, A., Kumar, R., Thober, S., Rakovec, O., Wanders, N., Zink, M., et al. (2018). Climate change alters low flows in Europe under global warming of 1.5, 2, and 3 °C. *Hydrology and Earth System Sciences*, *22*(2), 1017–1032. https://doi.org/10.5194/hess-22-1017-2018
- Milly, P., & Dunne, K. A. (2017). A hydrologic drying bias in water-resource impact analyses of anthropogenic climate change. JAWRA Journal of the American Water Resources Association, 53(4), 822–838. https://doi.org/10.1111/1752-1688.12538
- Milly, P. C., Dunne, K. A., & Vecchia, A. V. (2005). Global pattern of trends in streamflow and water availability in a changing climate. *Nature*, 438(7066), 347–350. https://doi.org/10.1038/nature04312
- Min, S.-K., Zhang, X., Zwiers, F. W., & Hegerl, G. C. (2011). Human contribution to more-intense precipitation extremes. Nature, 470(7334), 378–381. https://doi.org/10.1038/nature09763
- Morice, C. P., Kennedy, J. J., Rayner, N. A., & Jones, P. D. (2012). Quantifying uncertainties in global and regional temperature change using an ensemble of observational estimates: The HadCRUT4 data set. *Journal of Geophysical Research*, 117, D08101. https://doi.org/10.1029/ 2011JD017187
- Philip, S., Sparrow, S., Kew, S. F., van der Wiel, K., Wanders, N., Singh, R., et al. (2018). Attributing the 2017 bangladesh floods from meteorological and hydrological perspectives. *Hydrology and Earth System Sciences Discussions*, 2018, 1–32. https://doi.org/10.5194/ hess-2018-379
- Prudhomme, C., Giuntoli, I., Robinson, E. L., Clark, D. B., Arnell, N. W., Dankers, R., et al. (2014). Hydrological droughts in the 21st century, hotspots and uncertainties from a global multimodel ensemble experiment. *Proceedings of the National Academy of Sciences*, 111(9), 3262–3267. https://doi.org/10.1073/pnas.1222473110
- Smith, A., Sampson, C., & Bates, P. (2014). Regional flood frequency analysis at the global scale. Water Resources Research, 51, 539–553. https://doi.org/10.1002/2014WR015814
- Sousa, P., Trigo, R., Aizpurua, P., Nieto, R., Gimeno, L., & Garcia-Herrera, R. (2011). Trends and extremes of drought indices throughout the 20th century in the Mediterranean. *Natural Hazards and Earth System Sciences*, 11(1), 33–51. https://doi.org/10.5194/nhess-11-33-2011
- Sutanudjaja, E. H., van Beek, R., Wanders, N., Wada, Y., Bosmans, J. H. C., Drost, N., et al. (2018). PCR-GLOBWB 2: A 5 arcmin global hydrological and water resources model. *Geoscientific Model Development*, 11(6), 2429–2453. https://doi.org/10.5194/gmd-11-2429-2018
- Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93(4), 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1
- Tebaldi, C., & Knutti, R. (2007). The use of the multi-model ensemble in probabilistic climate projections. *Philosophical Transactions of the Royal Society of London A: Mathematical, Physical and Engineering Sciences, 365*(1857), 2053–2075. https://doi.org/10.1098/rsta.2007.2076
- Thober, S., Kumar, R., Wanders, N., Marx, A., Pan, M., Rakovec, O., et al. (2017). Multi-model ensemble projections of European river floods and high flows at 1.5, 2, and 3 degree global warming. *Environmental Research Letters*, 13, 014003. https://doi.org/10.1088/1748-9326/aa9e35
- United Nations Framework Convention on Climate Change (2015). Adoption of the Paris Agreement (*report No. FCCC/CP/2015/L.9/Rev.1*). United Nations. Retrieved from https://unfccc.int/resource/docs/2015/cop21/eng/l09r01.pdf
- Van der Wiel, K., Kapnick, S. B., Oldenborgh, G. J. v., Whan, K., Philip, S., Vecchi, G. A., et al. (2017). Rapid attribution of the August 2016 flood-inducing extreme precipitation in south Louisiana to climate change. *Hydrology and Earth System Sciences*, 21(2), 897–921. https://doi.org/10.5194/hess-21-897-2017
- Van der Wiel, K., Kapnick, S. B., Vecchi, G. A., Smith, J. A., Milly, P. C. D., & Jia, L. (2018). 100-year Lower Mississippi floods in a global climate model: Characteristics and future changes. *Journal of Hydrometeorology*, 19(10), 1547–1563. https://doi.org/10.1175/JHM-D-18-0018.1
- Vano, J. A., Nijssen, B., & Lettenmaier, D. P. (2015). Seasonal hydrologic responses to climate change in the Pacific Northwest. Water Resources Research, 51, 1959–1976. https://doi.org/10.1002/2014WR015909
- Wanders, N., & Van Lanen, H. A. J. (2015). Future discharge drought across climate regions around the world modelled with a synthetic hydrological modelling approach forced by three general circulation models. *Natural Hazards and Earth System Sciences*, 15(3), 487–504. https://doi.org/10.5194/nhess-15-487-2015
- Wanders, N., & Wada, Y. (2015a). Decadal predictability of river discharge with climate oscillations over the 20th and early 21st century. Geophysical Research Letters, 42, 10,689–10,695. https://doi.org/10.1002/2015GL066929

Wanders, N., & Wada, Y. (2015b). Human and climate impacts on the 21st century hydrological drought. *Journal of Hydrology*, 526, 208–220. https://doi.org/10.1016/j.jhydrol.2014.10.047

- Ward, P. J., Jongman, B., Aerts, J. C., Bates, P. D., Botzen, W. J., Loaiza, A. D., et al. (2017). A global framework for future costs and benefits of river-flood protection in urban areas. *Nature climate change*, 7(9), 642–646. https://doi.org/10.1038/nclimate3350
- Winsemius, H. C., Aerts, J. C. J. H., van Beek, L. P. H., Bierkens, M. F. P., Bouwman, A., Jongman, B., et al. (2015). Global drivers of future river flood risk. *Nature Climate Change*, 6(4), 381–385. https://doi.org/10.1038/nclimate2893
- Zotarelli, L., Dukes, M. D., Romero, C. C., Migliaccio, K. W., & Morgan, K. T. (2010). Step by step calculation of the Penman-Monteith evapotranspiration (FAO-56 method). Institute of Food and Agricultural Sciences. University of Florida.